BMJ Open Audit of data redaction practices in NICE technology appraisals from 1999 to 2019

Leeza Osipenko 💿

To cite: Osipenko L. Audit of data redaction practices in NICE technology appraisals from 1999 to 2019. *BMJ Open* 2021;**11**:e051812. doi:10.1136/ bmjopen-2021-051812

Prepublication history for this paper is available online. To view these files, please visit the journal online (http://dx.doi. org/10.1136/bmjopen-2021-051812).

Received 01 April 2021 Accepted 17 September 2021



© Author(s) (or their employer(s)) 2021. Re-use permitted under CC BY-NC. No commercial re-use. See rights and permissions. Published by BMJ.

Health Policy, London School of Economics and Political Science, London, UK

Correspondence to Dr Leeza Osipenko; I.osipenko@lse.ac.uk

ABSTRACT

Objectives To assess the extent and type of data redaction in all active technology appraisals (TA) and highly specialised technology (HST) evaluations issued by the National Institute for Health and Care Excellence (NICE) from its conception of the institute to September 2019. To propose policy recommendations for transparency. Methods Structured audit to establish extent of data redaction-proportion of appraisals and specific data categories and assess redaction by: indication, appraisal process, manufacturer, type of data-price, adverse events (AEs), clinical (excluding AEs), incremental qualityadjusted life-years. Longitudinal analysis over 20 years. Results All TAs with available documentation and active recommendations (n=408) and HSTs (n=10) published from March 2000 to 11 September 2019 have been assessed for data redaction. Overall, 333 TAs (81.6%) have data redaction, 86 (25.8%) of them are heavily redacted. Clinical data (excluding AEs) are redacted in 268 (65.7%) appraisals. AE data in 128 (31.4%), price in 238 (58.3%). In total, 87% of oncology appraisals have redacted data vs 78% of non-oncology appraisals. 91% of single TAs have redacted data vs 59% of multiple TAs. 25% of final guidance documents (e.g. Final Appraisal Determination - FAD) do not report one or more instance of clinical data. Data redaction increased substantially over time, and is currently at its highest level with 100% of TAs having at least some data redaction in 2019/2020, 96% of appraisals in 2018/2019% and 94% of appraisals in 2017/2018. All 10 HST evaluations have redacted data, with 4 of them being heavily redacted.

Conclusions Documents supporting NICE TA and HST recommendations are significantly redacted, thereby concealing clinical and economic data of importance to patients, clinicians and researchers. Documents remain redacted on the NICE website for years. Policy change is required to ensure transparency of data underpinning NICE's decisions.

INTRODUCTION

National Institute for Health and Care Excellence (NICE) was established in 1999. One of the key functions of NICE is to produce guidance on medicines, devices and other interventions for their use in the National Health Service (NHS). Before NICE issues any recommendations via the health technology assessment (HTA) process, a medicine or a

Strengths and limitations of this study

- This is the first comprehensive audit of data redaction practices in the technology appraisal programme at National Institute for Health and Care Excellence (NICE) over the 20-year period.
- This study establishes the extent of data redaction, type of redacted data and documents across different NICE processes.
- The findings can be externally validated by checking the original data made available by the author in the open-source repository.
- No systematic investigation has been performed to identify if academic-in-confidence data redacted from the NICE appraisal documentation have been published at a later date or are available through other sources, such as clinical study reports.

device must obtain a marketing authorisation from a regulatory agency: the European Medicines Agency (EMA), the UK's Medicines and Healthcare Regulatory Authority or accredited organisations issuing CE marks. The regulators review submissions by companies, which include laboratory methods, manufacturing specifications, preclinical and clinical data in order to make decisions on safety and efficacy of products. NICE accepts the decisions of the regulators and then evaluates further clinical and economic aspects of these products to ensure that all patients in the NHS have equitable access to the most clinically and cost-effective treatments that are available.

Technology appraisals (TA) is the core HTA programme that has been conducting multiple TAs (MTAs) since the inception of NICE. The single TA (STA) process was introduced in 2006 and gradually overtook MTAs. TAs predominantly focus on pharmaceuticals. Over time, other HTA programmes were introduced to evaluate interventional procedures, devices, diagnostic tests and medicines for ultrarare diseases—the programme for highly specialised technologies (HST). Each appraisal process is a rigorous exercise, which includes the evaluation of the manufacturer's submission by independent academic groups, public consultation and decision making by independent committees (not employed by NICE), whose conflicts of interest are openly managed.¹

The public, including doctors, patients and researches, has many potential routes to obtain detailed scientific information about new medicinal products and devices: regulatory agencies, HTA agencies, company websites, commercial and public data repositories, journal publications, etc. However, the availability, quality and visibility of such data is not guaranteed. Despite successes in trial reporting,^{2,3} publication of clinical study reports (CSRs) by regulators and industry,^{4,5} and data-sharing initiatives,⁶ access to data remains a problem due to resource constraints of the providers,^{4,7} commercial interests^{4,8} and publication bias.⁹

Industry-sponsored publications in scientific journals may not reflect issues and warnings issued by the regulators.¹⁰ Due to their format, many publications lack much of the detail available to the agencies¹¹ and are at a high risk of interpretation bias.⁹ The EMA started putting CSRs in the public domain in 2015⁵ and then stopped in 2018 due to workload constraints and lack of resources.¹² It remains suspended due to ongoing business continuity linked to the COVID-19 pandemic. The EMA is publishing clinical data for COVID-19 medicines in line with its exceptional transparency measures for treatments and vaccines for COVID-19¹³ Health Canada (HC) started to release CSRs in March 2019 and plans to make data available within 120 days after the agency's decision.⁴ The number of available reports remains small. The Food and Drug Administration initiated a pilot programme in 2018 to improve transparency to drugs evaluation and make CSRs available online.^{14 15} This initiative did not challenge the status quo. It has been decided that international harmonisation on data provision is required and until then data can be made available via individual requests.¹⁶ By contrast, NICE has and continues to enable access to all background documentation for currently active guidance. The processes and methods that NICE employs are structured and transparent. Every appraisal (unless replaced, terminated or withdrawn) has an online history page, which contains a comprehensive list of background documents on the technology and the process. These usually comprise of the following: the scoping document, list of consultees, manufacturer submission(s), consultees comments, assessment report(s) from the academic group, appeal documents, equality impact assessment, committee slides, appraisal consultation document and final appraisal determination (FAD). FAD is the output of the decision issued by NICE, which interprets the evidence and explains recommendations reached by the committee. This document is written in a clear language for the public, clinicians and patients to understand the underlying rationale for the decision. The final guidance is issued on the webpage for each TA summarising the recommendation, the technology, the

committee's discussion, research recommendations (if any), composition of the committee, implementation and additional information for the public.

Unfortunately, access to the documentation does not guarantee the visibility of data due to redaction practices. For example, GlaxoSmithKline has released CSRs for over 2500 studies; however, data redaction (especially relating to adverse events (AEs)) in these reports make them only moderately useful for researchers.⁴ Publication of heavily redacted clinical trial protocols remains a problem,^{17 18} in part because the level of redaction in CSRs submitted to the regulators is unknown, which has been criticised in the past.¹⁹ Bullement *et al* have shown that clinical data redaction in NICE STAs, which have confidential discounts (on price) via a patient access scheme (PAS)²⁰ is extremely high.²¹

Data redaction at NICE is sanctioned either as commercial-in-confidence (CiC) or academic-inconfidence (AiC). The NICE process guide defines CiC as 'information, the disclosure of which in public could have a significant impact on the commercial interests of a particular company²² ' and AiC as 'information, the disclosure of which in public would seriously jeopardise the ability of the data owner to publish the information in a scientific paper'.²² EMA has a policy on data redaction, according to which, clinical data are not considered to be commercially confidential information²³ besides a few exceptions limited to novel trial designs, and exploratory endpoints and biomarkers. NICE has an agreement with the Association of the British Pharmaceutical Industry (ABPI) on 'guidelines for the release of company data into the public domain during a health technology appraisal.²⁴ The title of the document implies that this agreement is applicable only to the actual appraisal process. However, the influence of this document spreads far beyond this remit, as many NICE documents remain redacted after the issuance of the guidance.²² The NICE TA process guide is reflective of this agreement and does not provide further clarity on data transparency once the appraisal process is completed and the recommendation is made public. It actually reinforces that 'a version (of manufacturer submission) for public release after the committee has met, in which all the confidential information is redacted, should be provided by the company' and that 'the data owner retains the right to make a final decision about the release of confidential information into the public domain.²³

Data held and released by the agencies differ in quantity and content. Regulatory agencies hold the most comprehensive sets of data for each product. Usually, these are highly technical and voluminous documents, and clinicians and patients may not have sufficient resources and knowledge to interpret these documents effectively.²⁵ Considering the rigour and transparency of the processes at NICE, the NICE website is an important source of evidence for the general public because: (1) background documents are available for each active guidance, (2) summaries of clinical and safety data focus on clinically relevant issues and (3) documents are written in an accessible language. Thus, NICE can play a critical role for evidence provision to the public in England and Wales. On a positive recommendation from NICE, medicines and devices become available to patients in the NHS. It would seem logical that at this stage (usually within 90 days), that all clinical data concerning the product are made publicly available.

Back in 2012, Strech and Littmann drew attention to the fact that the EMA and NICE restrict full access to unpublished evidence,²⁶ implying that the decision-making practices of the agencies have failed to account for different specifications of public interest. The authors requested the EMA and NICE to revise their policies on data transparency suggesting that the default restriction to data should be changed to a default access to data.²⁶ No action has been taken by NICE. Considering worrying findings on data redaction practices at NICE,²² this study aimed to conduct a complete audit of data redaction across all active NICE TAs and HSTs issued since the inception of the Institute 20 years ago, and establish the actual level and nature of data redaction practices at NICE.

METHODS

Creation of TA/HST ledger

An Excel workbook (see data repository²⁷) was used to create a ledger of all TAs and HSTs conducted by NICE between March 2000 (publication of TA1) and 11 September 2019 (publication of TA600). The status of each appraisal/evaluation was established from reviewing them individually on the NICE website (nice.org.uk). In the ledger, all TAs are arranged by the year of publication and classified by the status of the TA as of September 2019: recommended, in research, not recommended, replaced, terminated. The list of HSTs is presented separately.

Search strategy

All background documents for active (recommended, not recommended and in-research) TAs and HSTs were identified on the NICE website. These documents over the years follow varying formats but in general, have a consistent structure presenting information according to the NICE's process²² and methods²⁸ guides (which had a few editions since the establishment of the Institute) and evidence submission templates. For each appraisal, the following documents were searched for data redaction: FAD, assessment group (AG) or evidence review group (ERG) report (including erratum corrected documents and online supplemental analyses post consultation or submission of additional data by the company), manufacturer' submission (MS), and committee slides (if available). FADs were searched for the terms CiC and AiC. All reviewed documents were visually inspected for black boxes concealing text, tables and graphs as a means of redaction. In addition, word search for CiC and AiC was performed in each document.

Data extraction

Data extraction was performed by one person (the author) into a template.²⁷ For each appraisal, data were extracted on the basis of the following: TA number, sponsor company, year, process (STA)/MTA), technology name, technology type (pharmaceutical/non-pharmaceutical), indication, AiC and CiC (unrelated to price) data from FAD, redacted items (by type) from ERG/AG document, committee slides and MS. Appraisals from 2020/2021 were not included in the audit to extract specific data, however all of them were checked by the author for presence of data redaction to see if data redaction practice remains in place to date.

Assessing extent of redactions

In the NICE documentation data redaction can take two forms-either blacked out sections of the documents concealing the data (pertinent to background documents) or an explanation that data cannot be presented due to AiC or CiC designation (pertinent to FAD documents). To facilitate quantitative assessment of redactions in NICE appraisals, the following scoring system was adopted. Each appraisal was assessed for data redaction on: price, clinical data (excluding AEs), incremental quality adjusted life years (QALYs) and AEs. The score of zero was awarded to each item if no redactions were identified. The score of 1 was awarded to each item if (A) the appraisal referred in the FAD to confidential agreements and price discounts or if prices were redacted in any of the background documents, while the company did not have a commercial agreement or a PAS; (B) if any clinical data were redacted (eg, outcomes, results of indirect comparison analyses, utilities, baseline characteristics of patients in trials, time to treatment discontinuation, number of responders, modelled outcomes data); (C) if QALY gains were redacted and (D) if any data in the AE tables or paragraphs discussing AEs were redacted.

The aggregate score (0–3) was developed to quantify the level of redactions for the entire appraisal. Zero was awarded to appraisals with no redactions. Appraisals where redactions were minor—one paragraph or one item were noted (online supplemental table 12, please see reference²⁷) but no additional points for redaction were awarded. The score of 1 was awarded to those appraisals where either price or clinical data were redacted but not both. The score of 2 was awarded to appraisals: (A) where both price and clinical data were redacted; (B) where price were available, but clinical data redactions were substantial and/or FADs withheld clinical data under AiC or CiC. The score of 3 was awarded to heavily redacted appraisals (many items and pages redacted leaving little or no visible clinical data).

Data analysis

Data analysis (descriptive statistics in Excel) was performed by one researcher, the author. Data were analysed by the process type (STA, MTA, HST), by recommendation status (positive, negative and in research), by

Open access

technology type (pharmaceutical, non-pharmaceutical), by company (including only companies which products had 10 or more currently active STA recommendations). The proportion of FADs was calculated where clinical data were not reported due to CiC and AiC data nondisclosure. Longitudinal analysis was performed to assess the pattern of data redaction over the 20-year period. A separate analysis was undertaken to explore in more detail clinical and economic data redaction patterns. TAs and HSTs were identified that redacted the following data: (A) background characteristics of patients, (B) indirect comparisons (analysis and/or results), (C) economic model (D) clinical or modelled outcomes data, which is designated as CiC (rather than AiC), (E) ERG/AG comments/criticism and (F) clinical data redaction when price information is available.

Qualitative evaluation

A non-systematic approach was taken to present illustrative examples of data redaction practices from selected appraisals (n=25, online supplemental table 13, please see reference²⁷). EMA and HC CSRs were reviewed to assess data availability for the four appraisals, where NICE redacted data.

Validation

Validation of data extraction and analysis was performed by the researcher during the review of the data repository²⁷ to identify appraisals for the qualitative analysis. Each entry was rechecked for contents and completion of records. Additional searchers were performed where data were missing, and any errors and inconsistencies were corrected. All calculations were internally validated to ensure that the total number of appraisals in the analysed data set remained constant.

Patient and public involvement statement

Not applicable for this type of research.

RESULTS

As figure 1 shows, 191 (32%) appraisals were not reviewed because they were replaced (n=125) or terminated (n=66) and their documents were not available on the NICE website. A total of 409 (68%) TAs were reviewed, and TA77 (MTA) was excluded from the analysis because all documents pertinent to this appraisal were missing on the NICE website. Out of the remaining 408 reviewed TAs, 38 (9%) were negative (that is, they did not include any positive recommendations) and 370 (91%) were positive (or included at least one positive recommendation).

Across appraisals, reviewed manufacturer submissions range between 100 and 450 pages, ERG reports between 50 and 300 pages and AG reports between 80 and 450 pages. Out of all available appraisals/HST evaluations (n=418), 343 (82%) redacted data. Specifically, 238 (58.3%) TAs and 9 (90%) HSTs redacted price data, 268



1 http://prisma-statement.org/PRISMAStatement/FlowDiagram.aspx

Figure 1 Flow diagram of technology appraisals (TAs) and highly specialised technology evaluations (HSTs), which has been adapted from PRISMA to depict selection of TAs and HSTs for analysis. Briefly, a total of 600 TAs and 10 HSTs)were identified in the original data set. Records were screened to determine active appraisals and after removing terminated, missing and replaced appraisals, 408 TAs and 10 HSTs were included in the analysis. MTA, multiple technology appraisal; PRISMA, Preferred Reporting Items for Systematic Reviews and Meta-Analyses; STA, single technology appraisal.





Figure 2 Presence and extent of data redaction in technology appraisals (TAs) (n=408) and highly specialised technology evaluations (HSTs) (n=10) over 80% of appraisals had some data redaction. A total of 90% of STAs, 60% of MTAs and 100% of HSTs had some data redacted. In TAs at least some clinical data were redacted in over 65% of appraisals, QALYs in 35%, adverse events in over 30% and price in nearly 60%. A similar trend is shown for HSTs. AEs, adverse events; MTAs, multiple technology appraisals; QALY, quality-adjusted life-year.

(65.7%) TAs and all HSTs redacted at least some clinical data, 141 (34.6%) TA and 1 (10%) HST redacted incremental QALYs, 128 (31.4%) appraisals and 6 (60%) HSTs redacted AEs (figure 2). Overall, 86 (21.1%) TAs and 4 (40%) HSTs were heavily redacted (score of 3). For example, TA596 committee papers (page count—296, out of which over 60 pages are slides and organisational pages) have 89 pages with data redaction, many of these pages are blacked out completely.²⁹

The total number of positive oncology appraisals reviewed was 167. Of these, 146 (87%) had data redacted compared with 159 (78%) of non-oncology appraisals. In total, 59% of MTAs had data redactions compared with 91% of STAs. Six percent of MTAs were heavily redacted compared with 28% of STAs.

Data redaction increased dramatically over time (figures 3 and 4). At present, data redaction in TAs is at its highest level with 100% of TAs having at least some data redacted in 2019, 2020 and 2021. Ninety-six per cent of appraisals redacted data in 2018/2019 and 94% in 2017/2018. Over the past 10 years, STAs almost completely replaced the MTA process and since data redaction increased over the years this is likely to explain why STAs are more redacted than MTAs.

Figure 5 shows the extent of data redaction in the documents of the 11 companies that have ten or more currently active STA recommendations.

In non-recommended TAs (n=38), 28 (74%) had redacted data. Specifically, price was redacted in 16 (42%), clinical data in 22 (58%), QALYs in 11 (29%) and AEs in 9 (23.7%) appraisals. The total number of non-pharmaceutical recommended TAs reviewed was 25 (6.1% of total) and 14 (56%) of them had redacted data. Specifically, price was redacted in 7 (28%), at least some clinical data in 12 (48%), QALYs in 2 (8%) and AEs in 1 (4%) appraisal.

The total number of in-research appraisals reviewed was 34, where 32 appraisals were for medicines in the Cancer Drugs Fund (CDF) and 2 MTAs with no redacted data. All CDF appraisals had data redactions. Specifically, price was redacted in all CDF appraisals, at least some clinical data in 27 (84%), QALYs in 16 (50%) and AEs in 17 (53%) appraisals.

No black box redactions are used in any of the FADs; however, these documents refer to CiC and AiC statements to conceal information. Details on discounts and commercial agreements are CiC and are not reported in the FADs (besides a few exceptions of the complex



Figure 3 Redaction of data over time in NICE technology appraisals (n=408). This figure shows how data redaction has increased steadily and considerably over the past 8–10 years. In particular, price data have been redacted since 2008, in spikes that reflect over 50% of TAs by 2012. Accompanying this trend, quality-adjusted life-year (QALY) and adverse events data redaction have risen over the past decade. Importantly, at least some clinical data have been redacted in more than 50% of appraisals as early as 2005. NICE, National Institute for Health and Care Excellence; TAs, technology appraisals.

schemes).²⁰ Ninety-four (25%) of all FADs withheld at least some clinical data as CiC or AiC. Final recommendations presented on the webpage are usually reflective of the FADs, and for the FADs not reporting data under CiC or AiC no discrepancies were identified with the final recommendations.

Table 1 presents additional insights into data redaction practices across 408 TAs and 10 HSTs. The actual appraisal numbers for each category can be found in the data repository (please see reference²⁷).

Indirect comparisons were redacted in 88 (21.6%) appraisals. Baseline characteristics of patients in the trials were redacted in 51 (12.5%) appraisals, 28 (6.9%) appraisals redacted critical comments from ERG/AG and 41 (10%) appraisals redacted at least some clinical data as CiC rather than AiC. Out of 170 TAs that did not have price data redacted, 93 (54.7%) had clinical data redaction, 46 of these appraisals received aggregate scores of 2 or 3 indicating heavy redaction of clinical data.

Minimal redactions (eg, one paragraph) were identified in 34 (8.3%) appraisals.

Qualitative

To illustrate further the approach and extent of data redaction, a more detailed analysis was undertaken of selected appraisals published across different years. For example, TA199 redacted all R programming for the York economic analysis in addition to some clinical outcomes from the Impact-3 trial (change in Health Assessment Questionnaire Disability Index scores and stage 2 AEs).³⁰ TA303 has heavy data redaction across MS and ERG documents, but also the most redacted FAD with nine instances (in addition to price data) of undisclosed clinical data.³¹ Manufacturer submission for TA170 features redactions on 46 pages out of 136 pages.³² TA355 not only redacted clinical data and indirect comparison analysis and outputs, but also featured heavy redactions of the ERG critique of network meta-analysis, indirect comparisons, and summary of the trial characteristics.³³ AG report for TA59 has 136 instances of AiC data redaction.³⁴ TA161 has significant clinical data redaction, but also relies (along with TA160) on the fracture risks calculation by the AG, which uses the WHO algorithm that is AiC. Owing to this situation, the economic model for this appraisal was never released to consultees and commentators during this appraisal.³⁵ Besides TA303, none of these appraisals have a PAS and no logical explanation can be given to explain the egregious redaction of clinical data in TA303 to protect the confidentiality of the discount.

AEs and selective nature of redactions

The NICE website provides a high-level summary or some specific examples of adverse effects for each product. AEs data presented in the NICE background documents is taken from CSRs and usually provides complete summaries of AEs of all grades from the trials. This study finds



Figure 4 The pattern of data redaction in technology appraisals (n=408) at NICE. Over time, the amount and type of data redacted over the past 20 years has increased substantially. Initially, there were no data reactions in 2000/2001. In the first 6–7 years, only minimal price and clinical data were redacted. However, since 2008/2009 onwards, the amount of price and clinical data redaction has increased. The volume of data redaction has increased substantially since 2012. NICE, National Institute for Health and Care Excellence.

07-08

7

4 7

2

0 1

13 18

08-09

3

8

4

4

3

1

12

that in 31.4% of TAs and 60% of HSTs, these data are being redacted fully or partially. For example, TA485 MS provides detailed AEs data from the MOBILITY trial (different doses of sarilumab vs placebo), the MONARCH trial (adalimumab vs sarilumab) and the TARGET trial (different doses of sarilumab vs placebo), but all AEs from the ASCERTAIN trial (tocilizumab vs sarilumab) on pages 177-178 are redacted.³⁶ The ASCERTAIN trial reported safety and tolerability of subcutaneous sarilumab compared with intravenous tocilizumab in patients with rheumatoid arthritis was published in May 2019,³⁷ and reported no clinically meaningful differences in AEs between these two drugs. NHS patients and clinicians had no access to this information for 19 months and it is still not available on the NICE website. No CSRs for sarilumab are posted on the EMA or HC websites.^{38 39} There are many treatment options for patients with rheumatoid arthritis, however, patients for whom existing treatments do not work and who are seeking to switch to other products that may have a better AE profile, provision of such information could be critical.

6

No of appraisals/yr

70

60

50

40

30

20

10

0

no redacations

price + clinical

heavy redactions

price or clinical only

number of Appraisals/vr

00-01

3

0

0

0

3

02-03

4

0

0

4

03-04

6

1 1

2

1

9

04-05

3

1

1

7

05-06

2

0

0

0

2

06-07

6

3

2

0

11

01-02

2

1

1

0 0

4

TA292 (aripiprazole for bipolar disorder in adolescents) has a lot of clinical data redacted in background documents, but in the FAD, data on the incidence of clinically significant weight gain at 30 weeks was considered to be AiC by the manufacturer and was not presented, as well as data on the proportion of participants with a

body mass index on or above the 95th percentile at week 30.⁴⁰ This information would have been critical for clinicians and patients to know. This product has no PAS, but these data have never been made available on the NICE website. The CSRs for aripiprazole are not available on the EMA or HC websites.³⁸

The pattern of data redaction in the NICE documents is highly variable. From black boxes concealing the data, it is impossible to establish if redactions are AiC or CiC. For example, in TA462 the ERG report actually states on p400 of the committee papers: 'The ERG has taken a conservative approach and marked up, as AIC or CIC, any unmarked data whenever we were aware it was marked as AIC or CIC elsewhere in the submitted evidence.'41 Under AiC, presumably the entire data set from a study (all utility values or all AEs or all outcomes data) would not be made available; however, this is not necessarily the case. Data redaction can focus on specific parameters. For example, in TA395 (in table 34 MS, p76, please see reference⁴²) has half of the data in the table available on AEs and the other half the data are redacted; table 35 AE data are fully redacted (please see reference⁴²) but in table 36 (please see reference⁴²), two values of AEs are available and 11 are redacted.⁴² Furthermore, another three tables documenting AEs are fully redacted.⁴² These are the values from the same clinical trials. Another example is TA405 presenting selective

19-20

0

4

7

14

25



Figure 5 Data redaction in National Institute for Health and Care Excellence (NICE) technology appraisals (TAs) documents by 11 companies with 10 or more active single technology appraisals (STAs) recommendations (2006–2019). This figure shows data redaction practices for 11 companies which underwent many appraisals at NICE and had 10 or more active STAs at the time of the analysis. In almost all cases, price and some clinical data were redacted, and heavy data redactions are notable for all presented companies. Price and/or clinical data redactions were common for all companies. BMS, Bristol Myers Squibb.

redaction of AEs^{43} while these data are available on the EMA website³⁸ (figure 6).

Unredaction practices

Establishing whether data unredaction takes place was not possible because no versions of documents stating:

Table 1Additional insights into data redaction practices inTAs (n=408) and HSTs (n=10)							
Type of redaction	Appears in: (no of TAs/HSTs)						
Baseline (or subgroup) patient characteristics	51 (12.5%)						
Indirect comparisons	88 (21.6%)						
Economic model AiC or CiC	11 (2.7%)						
Clinical or modelled outcomes data designated as CiC	41 (10%)						
ERG/AG Comments/Criticism or Committee comments	28 (6.9%)						
Redacted clinical data when price data are available (n=170)	93 out of 170 (54.7%)						
Committee comments Redacted clinical data when price	93 out of 170 (54.7%)						

AG, assessment group; AiC, academic in confidence; CiC, commerical in confidence; ERG, evidence review group; HST, highly specialised technology; NMA, network meta-analysis; TA, technology appraisal.

'unredacted version' were identified. Possibly, unredacted data are presented in the TA163 ERG document in tables highlighted in yellow (pp16-17, 53–54)⁴⁴ and in TA426 (reconsideration of TA251) in different colour (pages 10–17, 28, 33 of the committee papers for TA251)⁴⁵; however, this TA features redacted data on indirect comparisons and AEs.⁴⁶

Some appraisals made attempts to indicate data availability, but no action was taken to make them available. The ERG report for TA298 states: 'please note that the ERG report was prepared before publication of the RADIANCE and REPAIR trials, therefore, some of the redacted information in this report is no longer confidential'.⁴⁷ This ERG document remains heavily redacted 8 years later. TA324 (2014) is a partial update of TA88, and UKPACE trial data are not presented and cited as CiC on pages 85-86,⁴⁸ despite the UKPACE trial being published in 2005.⁴⁹ Perhaps information for TA324 was copied from TA88 (2005). Data in relation to UKPACE trial in TA88 remain redacted (416 instances of CiC removed data).⁵⁰ In addition to specific data related to the trial, the documents are redacted for the AG comments on detection bias, on attrition bias, and on the statistical power of the UKPACE study. Such data should have been made available to researchers and clinicians helping them critically understand the quality of the UKPACE trial.

Data from manufacturer submission to NICE. TA405 Committee papers p 366 [42] Table 4.9: Comparison of adverse events in the RECOURSE trial and phase II trial (all grades)

(Based on Tables 41 and 43 of the CS1, the CSRs^{24, 25} and Mayer et al. 2015². Numbers extracted from the CS. Where the information was not reported in the CS or a discrepancy has been identified, relevant information from the CSR and/or Mayer et al. 2015 have been extracted as well.)

			Phas	e II	2 II			RECOURSE						
All grades AE	T	rifluridine/tipira	cil		BSC		Tı	Trifluridine/tipiracil			BSC			
	n	Number of events	%	n	Number of events	n	n	Number of events	%	n	Number of events	%		
Any event							533	524	98.3	265	247	93.2		
Any SAE	113	41* (21 patients)	18.6	57	8 (5 patients)	8.8	533	158	29.6	265	89	33.6		
Any treatment- related AE	113	109 ^{i*}	96.5	57	40 ⁺	70.2								
Nausea	113	73	64.6	57	16	28.1	533	258 [†]	48.4	265	63	23.8		
Vomiting	113	38	33.6	57	14	24.6	533	148^{\dagger}	27.8	265	38	14.3		
Decreased appetite							533	208†	39.0	265	78	29.4		
Diarrhoea	113	43	38.1	57	12	21.1	533	170^{\dagger}	31.9	265	33	12.5		
Abdominal pain [†]							533	113 ⁺	21.2	265	49	18.5		
Gastrointestinal disorders														
Neutropenia	113	81	71.7	57	1	1.8	528 [§]	358 Mayer: 353	67.8 Mayer: 67	263	2	0.8		
Leucopenia	113	86	76.1	57	2	3.5	528§	407	77.1	263	12	4.6		

Data from CSR report submitted to the EMA [37]

Study Protocol No.: J003-10040030

Table 12.2.3.1-1 Most Commonly Reported Adverse Events (>=10% of Patients in Either

fedDRA(ver.13.1)		TAS-102(N=113)		Placebo(N=57)
ystem Organ Class		Incidence	Rate*		Incidence	Rate*
Preferred Term	N	(27)	95% CI(%)	N	(%)	95%CI(%)
ny Events	111	(98.2)	[93.8, 99.8]	52	(91.2)	[80.7, 97.1]
strointestinal disorders	89	(78.8)	[70.1,85.9]	34	(59.6)	[45.8,72,4]
Abdominal distension	2	(1.8)	[0.2, 6.2]	6	(10.5)	[4.0,20-5]
Abdominal pain	22	(19.5)	[12.6, 28.0]	10	(17.5)	[8.7,29.9]
Diarrhoea	43	(38.1)	[29.1,47.7]	12	(21.1)	[01:4, 33.9]
Nausea	73	(64.6)	[55.0, 73.4]	16	(28.1)	6[17.0,41.5]
Stomatitis	17	(15.0)	[9.0, 23.0]	6	(10.5)	[4.0, 21.5]
Vomiting	38	(33.6)	[25.0, 43.1]	14	(24.6)	[14.1, 37.8]
eneral disorders and administration site conditions	84	(74.3)	[65.3,82.1]	27	(470)	[34.0, 61.0]
Fatigue	66	(58.4)	[48.8, 67.6]	24	(42.1)	[29.1, 55.9]
Influenza like illness	17	(15.0)	[9.0, 23.0]	1	2 (1.8)	[0.0, 9.4]
Oedema peripheral	14	(12.4)	[6.9, 19.9]	4 1	(7.0)	[1.9, 17.0]
Pyrexia	16	(14.2)	[8.3, 22.0]	25	(12.3)	[5.1,23.7]
fections and infestations	27	(23.9)	[16.4, 32.8]	290	(12.3)	[5.1, 23.7]
vestigations	108	(95.6)	[90.0, 98.5]	~40	(70.2)	[56.6, 81.6]
Alanine aminotransferase increased	10	(8.8)	[4.3, 15.7]	6 6	(10.5)	[4.0, 21.5]
Aspartate aminotransferase increased	23	(20.4)	[13.4, 29.0]	12	(21.1)	[11.4, 33.9]
Blood albumin decreased	29	(25.7)	[17.9, 34 P	11	(19.3)	[10.0, 31.9]
Blood bilirubin increased	33	(29.2)	[21.0,/38.5]	7	(12.3)	[5.1,23.7]
Blood lactate dehydrogenase increased	14	(12.4)	[6:9; 19.9]	13	(22.8)	[12.7, 35.8]
Blood sodium decreased	16	(14.2)	[3, 22.0]	4	(7.0)	[1.9, 17.0]
Haematocrit decreased	34	(30.1)	21.8, 39.4]	4	(7.0)	[1.9, 17.0]
Haemoglobin decreased	82	(32.6)	[63.4, 80.5]	9	(15.8)	[7.5,27.9]
Lymphocyte count decreased	39	(34.5)	[25.8, 44.0]	7	(12.3)	[5.1, 23.7]
Neutrophil count decreased	81	(7,57)	[62.4, 79.8]	1	(1.8)	[0.0, 9.4]
Platelet count decreased	40×	.(38.9)	[29.9, 48.6]	1	(1.8)	[0.0, 9.4]
Red blood cell count decreased	37	(32.7)	[24.2, 42.2]	2	(3.5)	[0.4, 12.1]
Weight decreased	© 23 , C	(20.4)	[13.4, 29.0]	1	(1.8)	[0.0, 9.4]
White blood cell count decreased	N 86 J	(76.1)	[67.2, 83.6]	2	(3.5)	[0.4, 12.1]
White blood cell count increased		(3.5)	[1.0, 8.8]	7	(12.3)	[5.1, 23.7]
Protein urine present	020	(17.7)	[11.2, 26.0]	6	(10.5)	[4.0, 21.5]
Blood alkaline phosphatase increased	17	(15.0)	[9.0.23.0]	15	(26.3)	[15.5, 39.7]
etabolism and nutrition disorders	2 71	(62.8)	[53.2, 71.7]	19	(33.3)	[21.4, 47.1]
Decreased appetite	70	(61.9)	[52.3, 70.9]	19	(33.3)	[21.4,47.1]
usculoskeletal and connective tissue disorders	35	(31.0)	[22.6, 40.4]	9	(15.8)	[7.5,27.9]
Back pain	12	(10.6)	[5.6, 17.8]	3	(5.3)	[1.1, 14.6]
rvous system disorders	27	(23.9)	[16.4, 32.8]	9	(15.8)	[7.5, 27.9]
chiatric disorders	5	(4.4)	[1.5, 10.0]	6	(10.5)	[4.0, 21.5]
spiratory, thoracic and mediastinal disorders	27	(23.9)	[16.4, 32.8]	6	(10.5)	[4.0, 21.5]
in and subcutaneous tissue disorders	22	(19.5)	[12.6, 28.0]	8	(14.0)	[6.3, 25.8]
Exfoliative rash	12	(10.6)	[5.6, 17.8]	5	(8.8)	[2.9, 19.3]

ot.

Figure 6 Trifluridine-tipiracil for previously treated metastatic colorectal cancer. Manufacturer submission to National Institute for Health and Care Excellence (NICE) versus European Medicines Agency (EMA) example showing redacted adverse events data from a phase II trial, from the manufacturer submission to NICE as part of TA405. This information is in stark contrast to the unredacted identical data available from the clinical study report submitted by the company to the EMA. CSR, clinical study reports; TA, technology appraisal.

There are examples of data redaction that clearly illustrate the lack of due diligence by NICE. TA450, where the committee papers⁵¹ are heavily redacted, was published on the NICE website on 28 June 2017. The TOWER trial results were published 3 months before on 2 March 2017 in the NEJM⁵² with all the supplemental material (detailed protocol, data on outcomes and AEs tables). The background documents for this appraisal remain redacted. Moreover, since many journal publications are behind publisher paywalls, access to the peer-review published literature is not freely available to patients and the public.

A heavily redacted TA592 was published on 9 August 2019.⁵³ All the redacted information related to clinical efficacy and the safety of cemiplimab is publicly available on the website of the Canadian regulator HC (figure 7). Documents related to this product on the HC website are redacted only for the personal patient data.³⁹ Canada released all the data on the 23 August 2019, which is only 2 weeks later than NICE. Another example (figure 8) relates to unredacted data on safety and efficacy of abemaciclib, which was posted on the HC website on 12 August 2019,³⁸ while a heavily redacted appraisal (TA579) by NICE was published on 9 May 2019.⁵⁴ For both appraisals, documents on the NICE website remain redacted.

CiC redaction of clinical data

In contrast to AiC, CiC data carries no obligation for even a theoretical future transparency. As this study shows, at least 41 (10%) FADs features CiC clinical data redaction (table 1) and only 41% of these appraisals, concealing clinical data as CiC, have commercial agreements with NICE. For example, TA34 HTA report has 67 'Commercial in confidence information removed' statements and 25 CiC redactions in tables.⁵⁵ TA337 background documents and the FAD⁵⁶ are full of CiC concealment of clinical data, which range from baseline characteristics of patients, to trial outcomes, to the treatment-related AEs, severe AEs and AEs that resulted in people stopping the study. In TA86, survival data are redacted (20 instances) with the reference to the unpublished study by Goss et al listing it as CiC.⁵⁷ Review documents from 2010 do not refer to the 'unpublished' Goss study. Further search for this publication did not yield any results, meaning that it has never been published. It is unclear why an unpublished study was classified as CiC and not AiC in the first instance. These appraisals have no PAS.

DISCUSSION

The NICE processes have been rightly praised for their high level of transparency.⁵⁸ The agency details names of the reviewers and conflicts of interest, types of evidence reviewed and evaluation steps and processes, meetings are held in public, and documentation is made available online. However, over the years, NICE has taken a lenient position on data redaction practices.

In total, 82% of NICE's documents pertinent to TAs and HSTs (100%) have varied levels of data redactions. Over

the past 20 years, data redaction has increased substantially. The volume of clinical data redaction is egregious. This study reveals that NICE does not unredact data even when availability of data in the public domain is obvious.

Policy implications

In Germany, it is not possible to redact data in the HTA process.⁵⁸ Both German agencies responsible for HTA, the IOWiG and G-BA, who are involved in the assessment report and appraisal documentation, respectively, have a strong legal basis for this transparency.⁵⁸ There is a law in place in Germany whereby the assessment of all new drugs that enter the German market (AMNOG, 2011) requires that a submission of complete data from the manufacturer and the information relating to the underlying decision making are in the public domain.^{59 60} In addition, full transparency of all information and data are required in the IQWiG's assessment report to enable full commenting procedure and decision-making transparent.⁵⁹ Importantly, strict penalties are issued in instances where incomplete data are submitted. Thus, any redaction by default would be considered incomplete data. This enhanced transparency in Germany has resulted in substantially more information being available on new drugs at the point of market entry.⁶⁰ The German example clearly illustrates that industry complies with transparency if public agencies enforce it. Germany is a bigger pharmaceutical market than the UK and traditionally characterised by an earlier launch of pharmaceutical products compared with the UK thus NICE can hardly argue that protection of industry's commercial interests jeopardises market access. In the current political climate in the UK, there are strong forces to make the system responsive to investment and the innovation agenda. The government's zeal to protect the industry's interests during the uncertainty posed by Brexit remains strong as corporations continue to pressure the UK government with threats of moving business outside the UK.⁶¹ In May 2019, the UK government has refused to sign up to a global resolution on greater transparency for drug pricing.⁶² The resolution urges governments and others buying health products to share information on actual prices paid, and pushes for greater transparency on patents, clinical trial results and other factors affecting pricing from laboratories to patients. The position of NICE (supported by the current legal framework) on redacting price data on drugs that have commercial agreements has been strongly favouring interests of the pharmaceutical industry since the establishment of the PAS in 2007.²⁰

Considering that in the NHS, medicines and most devices are provided free of charge, it could be argued that redaction of price data is irrelevant to the patients and taxpayers in England and Wales, but this is a weak argument for a number of reasons. First, in contrast to the French agency Haute Autorité de Santé and the Federal Joint Committee (G-BA) in Germany, NICE, does not explicitly state the judgement with regard to added Data from the manufacturer submission for TA592 page 503 Committee papers [52]

Table 1: Best Overall Tumor Response Rate by Independent central review, for All Patients with Opportunity for ≥3 Response Assessments on phase II study — Full Analysis Set

	mCSCC Cemiplimab: 3 mg/kg Q2W	IaCSCC Cemiplimab: 3 mg/kg Q2W	mCSCC Cemiplimab: 350 mg Q3W	Total
Bast Overall Turner	(N=59)	(N=64)	(N=44)	(N=167)
Best Overall Tumor Complete Response (CR) [a]	Response, n (%)			
Partial Response (PR) [a]				
Stable Disease (SD) [b]				
Non-CR/Non-PD [c]				
Progressive Disease (PD)				
Not Evaluable (NE) [d]				
Objective Response Rate				

Data from CSR report submitted to Health Canada. Submission control number 218718 [38]

Table 5: R2810-ONC-1540: Best Overall Tumor Response Rate by Independent Central Review, Group 1 and Group 2 – FAS

	mCSCC Cemiplimab	laCSCC Cemiplima	b:
*	3 mg/kg Q2W	3 mg/kg Q2W	Total
	^C (N=59)	(N=23)	(N=82)
Best Overall Tumor Response, n (%)	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	No de	
Complete Response (CR) [a]	4 (6.8%)	0 0 0	4 (4.9%)
Partial Response (PR) [a]	24 (40.7%)	10 (43.5%)	/34 (41.5%)
Stable Disease (SD) [b]	9 (15.3%)	9 (39.1%)	18 (22.0%)
Non-CR/Non-PD [c]	4 (6.8%)	0 6	4 (4,9%)
Progressive Disease (PD)	11 (18.6%)	2 (8.7%)	13 (15.9%)
Not Evaluable (NE) [d]	7 (11.9%)	2 (8.7%)	9 (11.0%)
Response Objective Response Rate (ORR: CR+PR)	28 (47,5%)	10 (43.5%)	38 (46.3%)
95% CI for ORR [e]	(34.3%, 60.9%)	(23.2%, 65.5%)	(35.3%, 57.7%)
99.99% CI for ORR [e]	(,,	(10.5%, 81.6%) [f]	0.
Complete Response Rate (CR) [a]	4 (6.8%)	0	4 (4.9%)
95% CI for CR Rate [e]	(1.9%, 16.5%)	(0.0%, 14.8%)	(1.3%, 12.0%)
Disease Control Rate (DCR.	41 (69.5%)	19 (82.6%)	60 (73.2%)
CR+PR+SD+Non-CR/Non-PD)			· (7)
95% CI for DCR [e]	(56.1%, 80.8%)	(61.2%, 95.0%)	(62.2%, 82.4%)
Durable DCR [g]	36 (61.0%)	16 (69.6%)	52 (63.4%)
95% CI for Durable DCR [e]	(47.4%, 73.5%)	(47.1%, 86.8%)	(52.0%, 73.8%)

Data cutoff as of 27 Oct 2017. Only patients who started treatment at least 9 months prior to the data cutoff date are included in Group 2 (laCSCC).

[a] CR/PR must be confirmed by repeated assessments no less than 4 weeks apart.

[b] SD criteria must be met at least once after a minimum duration of 39 days after first dose date.

Figure 7 Cemiplimab for treating metastatic or locally advanced cutaneous squamous cell carcinoma. Manufacturer submissions presented on the National Institute for Health and Care Excellence (NICE) website and on the Health Canada website (as of October 2019). Example of an entire table redaction of clinical tumour response outcomes from a manufacturer's submission to NICE as part of TA592. In contrast, the identical information is clearly presented in full (unredacted) in the clinical study report submitted by the manufacturer to Health Canada.

11

Data from manufacturer submission to NICE. TA579[53]

Table 18. Treatment-emergent adverse events by maximum CTCAE grade experienced by ≥10% of population of either arm of MONARCH 2,

		А	BE-FUL N=44	1			PI	BO-FUL N=2	23			
Ŀ	CTCAE Grade											
Preferred Term	Grade 1	Grade 2	Grade 3	Grade 4	All	Grade 1	Grade 2	Grade 3	Grade 4	All		
Patients with ≥1 TEAE, n (%)	-		241 (54.6)	26 (5.9)	435 (98.6)	-		46 (20.6)	5 (2.2)	199 (89.2		
Diarrhoea			59 (13.4)	0	381 (86.4)			1 (0.4)	0	55 (24.7)		
Neutropenia	Ŧ		104 (23.6)	13 (2.9)	203 (46.0)			3 (1.3)	1 (0.4)	9 (4.0)		
Nausea			12 (2.7)	NA	199 (45.1)			2 (0.9)	NA	51 (22.9)		
Fatigue	-		12 (2.7)	NA	176 (39.9)			1 (0.4)	NA	60 (26.9)		
Abdominal pain	-		11 (2.5)	0	156 (35.4)			2 (0.9)	0	35 (15.7)		
Anaemia	Ŧ		31 (7.0)	1 (0.2)	128 (29.0)			2 (0.9)	0	8 (3.6)		
Leukopenia	-		38 (8.6)	1 (0.2)	125 (28.3)			0	0	4 (1.8)		
Decreased appetite	-		5 (1.1)	0	117 (26.5)			1 (0.4)	0	27 (12.1)		
Vomiting	-		4 (0.9)	0	114 (25.9)			4 (1.8)	0	23 (10.3)		
Headache	-		3 (0.7)	NA	89 (20.2)			1 (0.4)	NA	34 (15.2)		
Dysgeusia	Ŧ		0	0	79 (17.9)			0	0	6 (2.7)		
Alopecia	T		NA	NA	69 (15.9)			NA	NA	4 (1.8)		
Thrombocytope nia	-		9 (2.0)	6 (1.4)	69 (15.6)			0	1 (0.4)	6 (2.7)		

Company evidence submission template for abemaciclib with fulvestrant for treating advanced hormone-receptor positive, HER2-negative breast cancer after endocrine therapy [ID1339]

© Eli Lilly and Company Limited (2018). All rights reserved

Page 77 of 181

Data from CSR report submitted to Health Canada. Submission control number 215268[38] Page 72 2.7.4. summary-clin-safety

9 (2.0)

2 (0.5)

3 (0.7)

17 (3.9)

0

0

3 (0.7)

10 (2.3)

4 (0.9)

19 (4.3)

17 (3.9)

10(2.3)

18 (4.1)

15 (3.4)

8 (1.8)

7 (1.6)

19 (4.3)

21 (4.8)

		0	<u>)</u> :		
Table	2	7	1	18	

Prefei Patien TEAE

> > Thrombocytopenia

Stomatitis

Cough

Pruritus

Dizziness

Constipation

ALT increased

AST increased

increased

Blood creatinine

35 (7.9)

48 (10.9)

47 (10.7)

23 (5.2)

44 (10.0)

49 (11.1)

45 (10.2)

25 (5.7)

27 (6.1)

Treatment-Emergent Adverse Events by Maximum CTCAE Grade Experienced by Greater Than or Equal to 10% of Population in Either Arm erred Term by Decreasing Frequency (All Grades) in the Abemaciclib Plus Fulvestrant Arm

4 (1.8)

18 (8.1)

26 (11.7)

5 (2.2)

21 (9.4)

12 (5,4)

11 (4.9)

7 (3.1)

1 (0,4)

1 (0.4)

5 (2.2)

3 (1.3)

3 (1.3)

4 (1.8)

1 (0.4)

2 (0.9)

2 (0.9)

0

0

0

1 (0.4)

4 (1.8)

0

0

0

6 (2.7)

0

Prefe
O. Safa

1	Safety Po	opulation									
	Same	and the second second	ib + Fulvest N=441	rant					Fulvestrant 223		
	CTCAE Grade										
Preferred Term	Grade 1	Grade 2	Grade 3	Grade 4	All	Grade 1	Grade 2	Grade 3	Grade 4	All	
Patients with ≥1 TEAE, n (%)	26 (5.9)	133 (30.2)	241 (54.6)	26 (5.9)	435 (98.6)	64 (28.7)	82 (36.8)	46 (20.6)	5 (2.2)	199 (89.2)	
Diarrhea	182 (41.3)	(140 (31.7)	59 (13.4)	· 0	381 (86.4)	43 (19.3)	11 (4.9)	1 (0.4)	0	55 (24.7)	
Neutropenia	23 (5.2)	63 (14.3)	104 (23.6)	13 (2.9)	203 (46.0)	4 (1.8)	1 (0.4)	3 (1.3)	1 (0.4)	9 (4.0)	
Nausea	129 (29.3)	58 (13.2)	12 (2.7)	NA	199 (45.1)	38 (17.0)	11 (4.9)	2 (0.9)	NA	51 (22.9)	
Fatigue	100 (22.7)	64 (14.5)	12 (2,7)	NA .	176 (39.9)	48 (21.5)	11 (4.9)	1 (0.4)	NA	60 (26.9)	
Abdominal pain	103 (23.4)	42 (9.5)	11 (2.5)	NA	156 (35,4)	24 (10.8)	9 (4.0)	2 (0.9)	NA	35 (15.7)	
Anemia	29 (6.6)	67 (15.2)	31 (7.0)	1 (0.2)	128 (29.0)	4 (1.8)	2 (0.9)	2 (0.9)	0	8 (3.6)	
Leukopenia	24 (5.4)	62 (14.1)	38 (8.6)	1(0.2)	125 (28.3)	/2 (0.9)	2 (0.9)	0	0	4 (1.8)	
Decreased appetite	69 (15.6)	43 (9.8)	5(1.1)	0	117 (26.5)	23 (10.3)	2 (0.9)	1 (0.4)	0	27 (12.1)	
Vomiting	79 (17.9)	31 (7.0)	4 (0.9)	0.0%	114 (25.9)	15 (6.7)	4 (1.8)	4 (1.8)	0	23 (10.3)	
Headache	62 (14.1)	24 (5.4)	3 (0.7)	NA	89 (20.2)	23 (10.3)	0 10 (4.5)	1 (0.4)	NA	34 (15.2)	
Dysgeusia	60 (13.6)	19 (4.3)	NA	NA	79 (17.9)	5 (2.2)	(0.4)	NA	NA	6 (2.7)	
Alopecia	60 (13.6)	9 (2.0)	NA	NA	69 (15,9)	4 (1.8)	0	NA	NA	4 (1.8)	

69 (15.6)

67 (15.2)

60 (13.6)

59 (13.4)

59 (13.4)

57 (12.9)

55 (12.5)

54 (12.2)

52 (11.8)

6(1.4)

0

0

1 (0.2)

NA

NA

NA

0

0

Figure 8 Abemaciclib with fulvestrant for treating hormone receptor-positive, HER2-negative advanced breast cancer after endocrine therapy: Manufacturer submissions posted on the National Institute for Health and Care Excellence (NICE) website and on the Health Canada website (as of October 2019). Example of adverse events data redaction from a manufacturer's submission to NICE as part of TA579. In contrast, the identical information is clearly presented (unredacted) in the clinical study report submitted by the manufacturer to Health Canada.

1 (0.4)

0

0

0

NA

NA

NA

0

0

6 (2.7)

23 (10.3)

30 (13.5)

12 (5 4)

25 (11.2)

13 (5.8)

13 (5.8)

15 (6.7)

1 (0.4)

therapeutic benefit (eg, major, minor, nothing new) of new medicines. This means that we have no way of seeing how the potential benefit has been quantified (some TAs may provide these data in technical documents but require training in health economics to interpret) and compared with existing medications in the same indication. Thus, if a confidentially agreed discount is high enough for the product to be deemed cost-effective according to the NICE methodology, regardless of the actual clinical benefit of this product, it will enter the NHS. This scenario represents a gaping missed opportunity to improve on the level of evidence relating to clinical benefit, which is especially warranted as much of the preclinical research leading to clinical trials is publicly funded.⁶³ Second, as Bullement et al illustrate, NICE's methodology actually requires redaction of some data in order to keep the price discounts confidential because the release of such data will allow the back-calculation of the price agreed.²¹ This methodological interdependence results in significant data redaction, especially in recent years where almost every recommended product has a commercial agreement. Third, the lack of pricing transparency in England has implications within and outside the country. Organisations such as ABPI propose that NICE is providing value for money to the NHS. In reality, this is a theoretical assumption rather than a fact⁶⁴ and taxpayers do not know whether they are paying more or less for medications than other European countries, how big the difference is, what services are being displaced to fund new medicines and devices, and whether modelled assumptions of clinical benefit on which NICE committees make decisions ever materialise in real life. In the meantime, the pharmaceutical industry continues to post top profitability margins,⁶⁵ and the clinical benefits of the growing number of pharmaceuticals entering the market remain questionable^{66 67} and might later be disappointing,⁶⁸ and healthcare system budgets are not coping with the pressure. International efforts should continue to bring price transparency on pharmaceuticals and devices across the globe, and the UK government can help lead this change.

There is no authority to hold the agency accountable to transparency. The legislative support of price and clinical data transparency at the national level will eliminate data redaction practices by NICE, which will serve not only NHS patients and clinicians, but a much wider clinical and research community around the world.

Context of other research

Research and inquiry into data redaction practices is limited. Goldacre in his book 'Bad Pharma' brings an example of data redaction in the NICE appraisal of ranibizumab stating that 'this level of censorship is not an everyday phenomenon'.¹⁹ As this study shows, substantial clinical data redaction is indeed an everyday practice, and it is the unredacted data that are becoming a transparency phenomenon. Recommendations that Strech and Littmann made to NICE on data transparency have not been adopted to date.²⁶ Since their publication, data redaction practices have become more prevalent. Bullement et al found that the volume of data censored as AiC in NICE appraisals was extremely high and called for the revision of the current practices. They also concluded that censoring appears to be performed on an ad hoc basis with no consistent pattern in the information censored.²¹ The findings of this study agree with those by Bullement et al, however, the authors are supportive of the AiC concept (with timely unredaction) while this work finds no grounds on which the AiC clinical data redaction should be practised. Panteli et al found the lack of transparency within HTA bodies on processes and consideration of data, suggesting data sharing between the agencies to improve the situation.⁶⁹ This research identified asymmetry in data visibility between regulatory agencies and NICE, and reiterates the need for interagency collaboration to maximise data transparency. Finally, EMA starting the release of CSRs in 2014, stated that it is guided by the conviction that public health interests must outweigh any private intellectual or commercial interest.⁵ Unfortunately, NICE is not following the German example and the call for implementation of similar legal requirements in other countries remains unanswered.^{59 60}

Limitations of the study

First, this study was performed by a single researcher, and while the results were internally validated by the researcher, external validation by an independent source was not possible. While this is a limitation of the study, all information has been extracted from resources in the public domain and is included in the online data repository,²⁷ which allows for external validation and re-assessment. Second, a subjective approach was used to determine an aggregate scoring system to assign levels of the extent and type of data redaction (0-3). However, the pragmatic methodology used to score redaction practices allows for replication. Finally, each manufacturer submission to NICE is accompanied by a checklist of confidential information, summarising the type and location of data redacted, but these checklists are not in the public domain and could not be used as an additional validation tool in this study.

Future research

Future work can analyse data redaction practices in regulatory and HTA agencies to enact policy change towards transparency and joint action to make data available. To unredact all NICE documents, publications that include data marked as AiC and CiC should be identified through searches or direct contact with the sponsors.

CONCLUSION

Data redaction in NICE appraisals is alarming in volume and context. Transparency of data is not regulated or sanctioned, and it undermines the role of NICE in protecting the health of patients and informing the public while

Open access

safeguarding the interests of industry. When clinical data are redacted by an HTA agency, patients, clinicians and researchers are deprived of information that might be crucial for their decision making with regard to a specific intervention. Policy change is required to ensure that all clinical data are fully available in the NICE documents for all the products for which NICE has issued guidance. All documents underpinning past decisions on the NICE website should be unredacted to make clinical data visible. International and harmonised policy changes are needed to enable global price transparency.

Acknowledgements LO thanks Dr Lisa Hutchinson for reviewing the manuscript; Dr Till Bruckner and Debra Winberg for filing Freedom of Information requests to NICE.

Contributors LO is the sole author of this project. LO came up with the original idea and methodology, conducted all data compilation, extraction, analysis and validation. She prepared all drafts of the manuscript and revisions.

Funding The authors have not declared a specific grant for this research from any funding agency in the public, commercial or not-for-profit sectors.

Competing interests None declared.

Patient and public involvement Patients and/or the public were not involved in the design, or conduct, or reporting, or dissemination plans of this research.

Patient consent for publication Not applicable.

Ethics approval Not required. This secondary desk research did not involve animals or human participants.

Provenance and peer review Not commissioned; externally peer reviewed.

Data availability statement Data are available in a public, open access repository. Data are available in a public, open access repository and can be used by anyone with no restriction-free access.Osipenko, Leeza. (2021). Ledger of data redactions in NICE Technology Appraisals (Data set). Zenodo. https://zenodo.org/record/5236080.

Open access This is an open access article distributed in accordance with the Creative Commons Attribution Non Commercial (CC BY-NC 4.0) license, which permits others to distribute, remix, adapt, build upon this work non-commercially, and license their derivative works on different terms, provided the original work is properly cited, appropriate credit is given, any changes made indicated, and the use is non-commercial. See: http://creativecommons.org/licenses/by-nc/4.0/.

ORCID iD

Leeza Osipenko http://orcid.org/0000-0002-7758-509X

REFERENCES

- 1 NICE. About us, 2021. Available: https://www.nice.org.uk/about/ what-we-do
- 2 TranspariMed & UAEM report. Clinical trial transparency at US universities, 2019. Available: https://www.transparimed.org/resources
- TranspariMed & UAEM. Clinical trial reporting by UK universities: progress report, 2019. Available: https://www.transparimed.org/ resources
- 4 UNDARK. Mantel, B Canada opens the door to public scrutiny of clinical drug trials, 2019. Available: https://undark.org/article/canadaopens-door-public-scrutiny-drug-trials/
- 5 Bonini S, Eichler H-G, Wathion Ň, et al. Transparency and the European medicines Agency — sharing of clinical trial data. N Engl J Med Overseas Ed 2014;371:2452–5.
- 6 Clinical study data Request platform, 2021. Available: https://www.clinicalstudydatarequest.com/
- 7 Brennan Z. Brexit impact: EMA suspends publication of clinical trial data, 2018. Available: https://www.raps.org/news-and-articles/newsarticles/2018/8/brexit-impact-ema-suspends-publication-of-clinica
- 8 Murray S. Eu Ombudsman calls for more transparency from AbbVie on Humira. PharmaLife, 2016. Available: http://www.pharmafile.com/ news/505079/eu-ombudsman-calls-more-transparency-abbviehumira

- 9 Sismondo S. Ghost-managed medicine: big pharma's invisible hands. Manchester, UK: Mattering Press, 2018.
- 10 Lurie P, Chahal HS, Sigelman DW, et al. Comparison of content of FDA letters not approving applications for new drugs and associated public announcements from sponsors: cross sectional study. BMJ 2015;350:h2758.
- 11 Wieseler B, McGauran N. Secrecy or transparency. CMAJ 2017;189:E185–6.
- 12 Doshi P. Ema scales back transparency initiatives because of workload. *BMJ* 2018;362:k3513.
- 13 EMA. Covid-19 latest updates, 2020. Available: https://www.ema. europa.eu/en/human-regulatory/overview/public-health-threats/ coronavirus-disease-covid-19
- 14 U.S. Food and Drug Administration. FDA's clinical data summary pilot program, 2018. Available: https://www.fda.gov/drugs/developmentapproval-process-drugs/fdas-clinical-data-summary-pilot-programquestions-frequently-asked-industry
- 15 Brennan Z. FDA plots shift away from Csr pilot to forge new transparency on drug approval process, 2019. Available: https:// www.raps.org/news-and-articles/news-articles/2019/6/fda-plotsshift-away-from-csr-pilot-to-forge-new-t
- 16 U.S. Food and Drug Administration. FDA continues to support transparency and collaboration in drug approval process as the clinical data summary pilot concludes, 2020. Available: https://www. fda.gov/news-events/press-announcements/fda-continues-supporttransparency-and-collaboration-drug-approval-process-clinical-datasummary
- 17 Wise J. BEACON trial in colorectal cancer is "single worst reported," says critic. BMJ 2019;367:I6057.
- 18 Marquardsen M, Ogden M, Gøtzsche PC. Redactions in protocols for drug trials: what industry sponsors concealed. J R Soc Med 2018;111:136–41.
- 19 Goldacre B. Bad pharma: how drug companies mislead doctors and harm patients. London: Fourth Estate, 2012.
- 20 NICE. Patient access schemes liaison unit, 2021. Available: https:// www.nice.org.uk/about/what-we-do/patient-access-schemesliaison-unit
- 21 Bullement A, Taylor M, McMordie ST, et al. Nice, in confidence: an assessment of Redaction to obscure confidential information in single technology appraisals by the National Institute for health and care excellence. *Pharmacoeconomics* 2019;37:1383–90.
- 22 NICE. Guide to the processes of technology appraisal, 2014. Available: https://www.nice.org.uk/process/pmg19
- 23 HMA and EMA. HMA/EMA guidance document on the identification of commercially confidential information and personal data, 2012. Available: https://www.ema.europa.eu/en/documents/other/ heads-medicines-agencies/european-medicines-agency-guidancedocument-identification-commercially-confidential-information_en. pdf
- 24 ABPI and NICE. Agreement between the association of the British pharmaceutical industry (ABPI) and the National Institute for health and clinical excellence (NICE) on guidelines for the release of company data into the public domain during a health technology appraisal, 2011. Available: https://www.nice.org.uk/Media/Default/ About/what-we-do/NICE-guidance/NICE-technology-appraisals/ Guidelines-for-the-release-of-company-data-into-the%20publicdomain-during-a-health-technology-appraisal.pdf
- 25 Way D, Bouder F, Löfstedt R, et al. Medicines transparency at the European medicines Agency (EMA) in the new information age: the perspectives of patients. J Risk Res 2016;19:1185–215.
- 26 Strech D, Littmann J, proportionality Lof. Lack of proportionality. seven specifications of public interest that override post-approval commercial interests on limited access to clinical data. *Trials* 2012;13:100.
- 27 Osipenko L. Ledger of data redactions in NICE Technology appraisals [Data set]. Zenodo, 2021. Available: https://zenodo.org/ record/5236080
- 28 NICE. Guide to the methods of technology appraisal 2013, 2013. Available: https://www.nice.org.uk/process/pmg9/chapter/foreword
- 29 NICE. Committee papers. TA 596. Risankizumab for treating moderate to severe plaque psoriasis, 2019. Available: https://www. nice.org.uk/guidance/ta596/documents/committee-papers
- 30 CRD/CHE Technology Assessment Group, University of York. TA199. etanercept, infliximab and adalimumab for the treatment of psoriatic arthritis: a systematic review and economic evaluation. technology assessment report for NICE MTA, 2010. Available: https://www.nice.org.uk/guidance/ta199/documents/ psoriatic-arthritis-etanercept-infliximab-adalimumab-review-acdassessment-report3
- 31 NICE. Committee papers. TA 303. teriflunomide for the treatment of relapsing forms of multiple sclerosis, 2013. Available: https://www.

<u>ð</u>

Open access

nice.org.uk/guidance/ta303/documents/multiple-sclerosis-relapsing-teriflunomide-evaluation-report4

- 32 NICE. TA 170. Bayer submission. rivaroxaban (Xarelto®) for the prevention of venous thromboemobolism (VTe) in adult patients undergoing elective hip or knee replacement surgery, 2008. Available: https://www.nice.org.uk/guidance/ta170/documents/bayer-healthcare2
- 33 NICE. Committee papers. TA 355. edoxaban tosylate for preventing stroke and systemic embolism in people with non-valvular atrial fibrillation, 2015. Available: https://www.nice.org.uk/guidance/ta355/ documents/atrial-fibrillation-nonvalvular-edoxaban-tosylate-id624committee-papers2
- 34 The School of Health and Related Research (ScHARR). University of Sheffield TA59 electroconvulsive therapy (ECT) for depressive illness, schizophrenia, catatonia and mania. Technology assessment report for NICE MTA, 2002. Available: https://www.nice.org.uk/guidance/ ta59/documents/final-assessment-report-electroconvulsive-therapyect-for-depressive-illness-schizophrenia-catatonia-and-mania2
- 35 NICE. Final appraisal determination for TA161 alendronate, etidronate, risedronate, raloxifene, strontium ranelate and teriparatide for the secondary prevention of osteoporotic fragility fractures in postmenopausal women, 2010. Available: https://www.nice.org.uk/ guidance/ta161/documents/osteoporosis-secondary-preventionincluding-strontium-ranelate-final-appraisal-determination-document2
- 36 NICE. Committee papers. TA485. Sarilumab for previously treated moderate to severe active rheumatoid arthritis, 2017. Available: https://www.nice.org.uk/guidance/ta485/documents/committee-papers
- 37 Emery P, Rondon J, Parrino J, et al. Safety and tolerability of subcutaneous sarilumab and intravenous tocilizumab in patients with rheumatoid arthritis. *Rheumatology* 2019;58:849–58.
- 38 European Medicines Agency. EMA clinical data portal, 2021. Available: https://clinicaldata.ema.europa.eu/
- 39 Government of Canada. Health Canada clinical data portal, 2021. Available: https://clinical-information.canada.ca/
- 40 NICE. Final appraisal determination for TA292 aripiprazole for treating moderate to severe manic episodes in adolescents with bipolar I disorder, 2013. Available: https://www.nice.org.uk/guidance/ta292/ documents/bipolar-disorder-children-aripirazole-final-appraisaldetermintation-document2
- 41 NICE. Committee papers. TA462. nivolumab for treating relapsed or refractory classical Hodgkin lymphoma, 2017. Available: https:// www.nice.org.uk/guidance/ta462/documents/committee-papers
- 42 NICE. Committee papers. TA395. Ceritinib for previously treated anaplasticlymphoma- kinase-positive non-small-cell lung cancer, 2015. Available: https://www.nice.org.uk/guidance/ta395/ documents/lung-cancer-nonsmallcell-anaplastic-lymphoma-kinasepositive-previously-treated-ceritinib-id729-committee-papers-2
- 43 NICE. Committee papers. TA 405 Trifluridine–tipiracil for previously treated metastatic colorectal cancer. P366, 2016. Available: https://www.nice.org.uk/guidance/ta405/documents/committee-papers
- 44 NICE, West Midlands HTA Collaboration, U of Birmingham. Infliximab for the treatment of acute exacerbations of ulcerative colitis. TA163, 2008. Available: https://www.nice.org.uk/guidance/ta163/ documents/evaluation-report-evidence-review-group-report2
- 45 NICE. Committee papers. TA 426. dasatinib, nilotinib and standarddose imatinib for the first line treatment of chronic myeloid leukaemia, 2011. Available: https://www.nice.org.uk/guidance/ta426/ documents/committee-papers-2
- 46 NICE. Committee papers. TA 426. CDF rapid review. dasatinib for the first-line treatment of chronic myeloid leukaemia (part review of TA251), 2016. Available: https://www.nice.org.uk/guidance/ta426/ documents/committee-papers
- 47 NICE. Committee papers. TA 298. ranibizumab for treating choroidal neovascularisation associated with pathological myopia, 2013. Available: https://www.nice.org.uk/guidance/ta298/documents/ choroidal-neovascularisation-pathological-myopia-ranibizumabevaluation-report2
- 48 NICE, BMJ-TAG. Dual-Chamber pacemakers for treating symptomatic bradycardia due to sick sinus syndrome without atrioventricular block, part review of technology appraisal 88. HTa report TA 324, 2014. Available: https://www.nice.org.uk/guidance/ ta324/documents/bradycardia-without-atrioventricular-block-dualchamber-pacemakers-part-review-of-ta88-assessment-report2
- 49 Toff WD, Camm AJ, Skehan JD, et al. Single-chamber versus dualchamber pacing for high-grade atrioventricular block. N Engl J Med 2005;353:145–55.

- 50 NICE, PenTAG, Universities of Exeter & Plymouth. Effectiveness and cost effectiveness of dual vs. single chamber pacemakers. HTA report 2004, 2004. Available: https://www.nice.org.uk/guidance/ ta88/documents/assessment-report-for-appraisal-of-dual-chamberpacing4
- 51 NICE. Committee papers. TA450. Blinatumomab for treating Philadelphia chromosome-negative relapsed or refractory acute lymphoblastic leukaemia, 2016. Available: https://www.nice.org.uk/ guidance/ta450/documents/committee-papers
- 52 Kantarjian H, Stein A, Gökbuget N, et al. Blinatumomab versus chemotherapy for advanced acute lymphoblastic leukemia. N Engl J Med 2017;376:836–47.
- 53 NICE. Committee slides. TA592. Cemiplimab for treating metastatic or locally advanced cutaneous squamous cell carcinoma, 2019. Available: https://www.nice.org.uk/guidance/ta592/documents/1
- 54 NICE. Committee papers. TA 579. Abemaciclib with fulvestrant for treating hormone receptor-positive, HER2-negative advanced breast cancer after endocrine therapy, 2019. Available: https://www.nice. org.uk/guidance/ta579/documents/committee-papers
- 55 NICE. Nhs centre for reviews and dissemination. HTA report: a rapid and systematic review of the clinical effectiveness and cost-effectiveness of trastuzumab for breast cancer. TA34, 2001. Available: https://www.nice.org.uk/guidance/ta34/documents/ assessment-report-on-trastuzumab-for-advanced-breastcancer2
- 56 NICE. Final appraisal determination for TA337. rifaximin for preventing episodes of overt hepatic encephalopathy, 2015. Available: https://www.nice.org.uk/guidance/ta337/documents/ rifaximin-for-preventing-episodes-of-overt-hepatic-encephalopathyfinal-appraisal-determination-document2
- 57 NICE, West Midlands HTA Collaboration, U of Birmingham. Imatinib for the treatment of patients with unresectable and/or metastatic gastro-intestinal stromal tumours – a systematic review and economic evaluation, 2003. Available: https://www.nice.org. uk/guidance/ta86/documents/gastrointestinal-stromal-tumoursimatinib-assessment-report-2
- 58 Bruckner T. Transparency in health technology assessment: NICE, IQWIG and HAS in comparison. March 2020 by HealthWatch, 2020. Available: https://www.healthwatch-uk.org/HTA.pdf
- 59 IQWIG. Early benefit assessment increases transparency for study data, 2015. Available: https://www.iqwig.de/en/presse/press-releases/press-releases-detailpage_10643.html
- 60 Köhler M, Haag S, Biester K, et al. Information on new drugs at market entry: retrospective analysis of health technology assessment reports versus regulatory reports, Journal publications, and registry reports. *BMJ* 2015;350:h796.
- 61 Kollewe J, Scruton P. What are Brexit contingency plans for pharmaceutical industry? The Guardian, 2019. Available: https:// www.theguardian.com/politics/2019/feb/19/what-are-brexitcontingency-plans-for-pharmaceutical-industry
- 62 Lamble L. UK refuses to back 'game-changing' resolution on drug pricing. The Guardian, 2019. Available: https://www.theguardian. com/global-development/2019/may/28/uk-refuses-to-back-game-changing-resolution-on-drug-pricing
- 63 Galkina Cleary E, Beierlein JM, Khanuja NS, et al. Contribution of NIH funding to new drug approvals 2010-2016. Proc Natl Acad Sci U S A 2018;115:2329–34.
- 64 Claxton K, Sculpher M, Palmer S, et al. Causes for concern: is NICE failing to uphold its responsibilities to all NHS patients? *Health Econ* 2015;24:1–7.
- 65 Sherman E. Drug companies have Big R&D expenses and still make large profits, 2019. Available: https://fortune.com/2019/03/01/drug-companies-rd-profits/
- 66 Wieseler B, McGauran N, Kaiser T. New drugs: where did we go wrong and what can we do better? *BMJ* 2019;366:I4340.
- 67 Prescrire International. Drugs in 2018: a brief review. *Rev Prescrire* 2019;39:142–4.
- 68 Davis C, Naci H, Gurpinar E, et al. Availability of evidence of benefits on overall survival and quality of life of cancer drugs Approved by European medicines agency: retrospective cohort study of drug approvals 2009-13. *BMJ* 2017;359:j4530.
- 69 Panteli D, Nolting A, Eckhardt H, et al. Published and unpublished evidence in coverage decision-making for pharmaceuticals in Europe: existing approaches and way forward. *Health Res Policy* Syst 2016;14:6.